Behind the Scenes: The Corporate Governance Preferences of Institutional Investors

JOSEPH A. McCAHERY, ZACHARIAS SAUTNER, and LAURA T. STARKS*

February 2015

ABSTRACT

We survey institutional investors to better understand their role in the corporate governance of firms. We document widespread behind-the-scenes intervention and governance-motivated exit. Both governance mechanisms are viewed as complementary devices, in which intervention typically occurs prior to an eventual exit. We find that long-term investors and investors that are less concerned about stock liquidity intervene more intensively. Finally, most investors use proxy advisors and believe that their information improves their own voting decisions.

^{*}Joseph McCahery is at Tilburg University. Zacharias Sautner is at Frankfurt School of Finance & Management. Laura T. Starks is at McCombs School of Business, University of Texas at Austin. We are very grateful to Campbell Harvey, John Graham, Michael Roberts, an Associate Editor, anonymous referees, those who helped distribute our survey, and most of all, to the anonymous respondents to the survey. We would also like to thank Donna Anderson, Alon Brav, Marco Becht, Stephen Brown, Mark Coffeldt, Jonathan Feigelson, Merrit Fox, Stuart Gillan, Allaudeen Hameed, Bram Hendricks, Paulus Ingram, Jacqueline Johnson, Sanford Leeds, Amir Licht, Florencio Lopez-de-Silanes, Enrico Perotti, Randall Thomas, and David Yermack for additional help with the survey design and delivery. We further benefitted from the comments of Renee Adams, Ines Chaieb, Steve Choi, Amil Dasgupta, Alex Edmans, Rüdiger Fahlenbrach, Miguel Ferreira, Marc Gabarro, Mariassunta Giannetti, Kai Li, Florencio Lopez-de-Silanes, Ernst Maug, Alexander Nagornov, Enrico Perotti, Luc Renneboog, Armin Schwienbacher, Rene Stulz, Randall Thomas, Jaap Winter, and seminar participants at the AFA Meetings, EFA Meetings, FMA Asia, FIRS, Harvard Conference on Corporate Governance, the Conferences on Innovation and Business Law, the Rotterdam Workshop on Executive Compensation and Corporate Governance, the Vanderbilt/Amsterdam Conference in Cagliari, the LUISS/Tilburg Conference in Rome, WU Endowment Conference, Ohio State University, Seoul National University, ETH Zurich, University of Melbourne, University of Amsterdam, and University of Queensland.

Theoretical and empirical research on corporate governance makes assumptions and draws inferences regarding the role of institutional investors. Yet, we have little direct knowledge about how institutional investors engage with portfolio companies because many interactions occur behind the scenes. That is, unless institutions publicly express their approval or disapproval of a firm's activities or management, little is known about their preferences and private engagements with portfolio firms. Our goal in this paper is to rectify this knowledge gap by conducting a survey among institutional investors.

As early as Hirschman (1970), researchers have highlighted the two active choices that institutional investors have when they become unhappy with a portfolio firm: (i) they can engage with management to try to institute change ("voice" or direct intervention); or (ii) they can leave the firm by selling shares ("exit" or "voting with their feet"). Subsequently, theoretical models have documented the governance benefits of corrective actions through voice.² These theories have recently been complemented by models showing that the threat of exit can also discipline management.³ This raises the question of whether institutional investors, in response to dissatisfaction with portfolio firms, take actions that support the validity of these theories.

The 143 respondents to our survey, mostly very large institutional investors with a long-term focus, indicate that voice, especially when conducted behind the scenes, is highly important. For example, 63% of the respondents state that, in the past five years, they have engaged in direct discussions with management, and 45% have had private discussions with a company's board *outside* of management's presence. In addition, we find evidence that investor horizon makes a difference. First, long-term investors intervene more intensively than short-term investors. Second, investors who choose engagement do so more often because of concerns over a firm's corporate governance or strategy than over short-term issues. Both

¹ For some recent anecdotal evidence on behind-the-scenes activism, see Burr (2012).

² See, for example, Shleifer and Vishny (1986), Huddart (1993), Admati, Pfleiderer, and Zechner (1994), Maug (1998), Kahn and Winton (1998), Bolton and von Thadden (1998), and Faure-Grimaud and Gromb (2004).

³ See Admati and Pfleiderer (2009), Edmans (2009), and Edmans and Manso (2011).

⁴ In the survey we explicitly asked about *engagement* through discussions, rather than casual conversations at, for example, investor relations events.

findings support the view that interventions are not primarily conducted by short-term, myopic activists who intend solely to reap short-term gains (e.g., Bebchuk, Brav, and Jiang (2014)).

Nevertheless, the institutional investors also indicate that they face impediments to their activism, with the most important hurdles being free rider problems (as in Shleifer and Vishny (1986)) and legal concerns over "acting in concert" rules. Our results also show that investors who are more concerned about liquidity (and hence probably hold more liquid stocks) use voice less intensively. This is consistent with Coffee (1991), Bhide (1993), and Back, Li, and Ljungqvist (2014), who argue that market liquidity discourages intervention.

A central challenge arises in testing whether institutional investors use the threat of exit and whether it is effective in bringing about changes in management behavior. The challenge is that the *threat* of exit is, by definition, unobservable. In fact, if the threat is credible, exit itself does not take place. Our survey sheds light on the exit mechanism because we can ask institutions (i) whether they use exit as a governance device; and (ii) whether they believe that the exit threat is an effective disciplinary device.

The investors in our survey view exit as a viable strategy with 49% (39%) stating that they had exited because of dissatisfaction with performance (governance) in the past five years. Based on their experience with portfolio firms, 42% of the investors believe that the exit threat is effective in disciplining management. As our respondents tend to be dedicated, long-term investors who engage privately, it is plausible that their potential exit truly is a threat. The investors further indicate that exit is a complement to a voice strategy rather than a substitute, and intervention typically occurs prior to an eventual exit. The survey results further suggest that the effectiveness of the exit threat depends on investors' equity stake size, whether other investors also exit for the same reason, managerial equity ownership, and whether other large shareholders are also present.

Finally, we consider an increasingly controversial role in corporate governance: the role of proxy advisors.⁵ Although the respondents raise some concerns over conflicts of interest in proxy advisory firms, arising from the offering of consulting services, they find their advice to be of value. 60% of the respondents use proxy advisors, and about half of these respondents actually use the services of more than one advisor. The investors report that proxy advisors help them make better voting decisions, but that they remain their own decision makers. Further, investors that use proxy advisors engage their portfolio companies more intensively through voice, rather than substituting the proxy advice for their own voice. Contrary to some regulatory and media beliefs, this suggests that the use of proxy advisors does not necessarily imply that investors take a passive governance role. These results are in line with Aggarwal, Erel, and Starks (2014) and Iliev and Lowry (2015), who find that proxy voters do not uniformly follow the recommendations of proxy advisors.

We contribute to the literature by providing direct evidence of institutional investors' preferences and actions. The few related studies also using direct evidence have examined activism either by one specific investor or by hedge funds. The studies that focus on one activist show that private interventions are often successful. Given that each of these studies focuses on a single institution, the extent to which this evidence can be generalized is unclear. The studies on hedge funds show that their activism is met with abnormal target returns, leads to more innovation and higher performance, and even disciplines non-targeted firms. Although this evidence is important, hedge funds are one type of investor that has particularly strong incentives to engage, can take more concentrated positions, and faces fewer conflicts of

⁵ This role has been assailed by institutional investors, regulators and politicians. SEC Commissioner Daniel M. Gallagher, for example, states that he has "...grave concerns as to whether investment advisers are indeed truly fulfilling their fiduciary duties when they rely on and follow recommendations from proxy advisory firms" (Gallagher (2013)). In 2014 the SEC issued guidance on the use of proxy advisors and the European Commission drafted a proposal for a Directive with the goal of enhancing the reliability and quality of advice by proxy advisors.

⁶ For example, Smith (1996) studies CalPERS, Carleton, Nelson, and Weisbach (1998) study TIAA-CREF, Becht et al. (2009) study the Hermes UK Focus Fund, and Dimson, Karakas, and Li (2014) study another U.K. activist.

⁷ See Brav et al. (2008a), Brav, Jiang, and Kim (2015), Bebchuk, Brav, and Jiang (2014), Brav et al. (2014), and Gantchev, Gredil, and Jotikasthira (2014).

interest (Brav, Jiang, and Kim (2010)). Thus, our evidence is important in providing a more generalizable view of institutional investor motives and activities. The widespread use of behind-the-scenes engagement we document highlights the importance of our survey approach and suggests that many investors might be more active than would be inferred from observational data only.

Section I describes the survey. Section II presents results on engagement channels. Section III studies the exit threat. Section IV presents results on impediment and triggers of engagement, Section V reports results on proxy advisors, and Section VI concludes.

I. Survey Design

A. Survey Development and Delivery

We developed our survey by considering questions that would provide insights into outstanding issues in research on shareholder engagement. We used both an online and a paper version of the survey. The survey questions are provided in Internet Appendix IA.I, and details on the survey design are in Internet Appendix IA.II.

We used three delivery channels. First, we distributed the paper version of the survey at three conferences: (1) the International Corporate Governance Network (ICGN) Dinner Debate in Rotterdam in December 2012, which had about 40 participants, from which we received 10 responses; (2) the ICGN Annual Event in New York in June 2013, from which we received 26 responses⁸; and (3) the Istanbul Pension Funds Conference in July 2013, which had about 50 participants, from which we received 8 responses.

Second, we sent invitations to participate in the online survey in a targeted emailing in July 2013. A very large and governance-sensitive institutional investor provided us with contact details for 54 CIOs or senior corporate governance specialists at other large institutional investors. We sent personal emails to these contacts, inviting them to participate in the survey.

4

⁸ Although this conference had participants in the hundreds, many of the participants were not institutional investors and thus could not complete the survey.

We also sent emails to about 50 personal contacts of the authors who worked at different institutional investors. From these two sets of emails we received 67 responses.

Third, using the FactSet database, we sent invitations to participate in the online survey to approximately 3,000 portfolio managers and received 32 responses.

In total we had 143 responses to all invitations, implying a response rate of about 4.3% across the three delivery channels and relative to the approximately 3,300 invitations. This response rate is a little lower than that of some other surveys in finance (e.g., 5.3% in Brav et al. (2008b); 5.4% in Dichev et al. (2013)). However, it is approximately equal to the 4.5% response rate in the quarterly CFO survey by Duke University.⁹

In a survey of the opinions of economic agents such as this one, we naturally face the risk that respondents answer in a strategic or untruthful fashion. To mitigate these concerns, we conducted the survey anonymously and did not require (or ask) respondents to reveal their names or employers. We further emphasized that individual responses would be treated as confidential. Conversations with a number of respondents also indicated that they would not spend time filling out the survey if they intended to answer untruthfully.

B. Respondent Characteristics

Table I provides summary statistics on the respondents. As not all respondents provided information on investor or investment characteristics, the number of observations falls below the 143 total. Table I, Panel A shows that, given their position, our respondents should be very knowledgeable about their firms' preferences and actions with regard to shareholder engagement. The largest numbers of respondents are either corporate governance experts (29%) or portfolio managers (27%). Moreover, about one-third of the respondents are very senior in their organizations, being either CIOs (18%) or board members (14%). The large

⁹ The low response rate is primarily the result of the mass emailing using the FactSet database. If we exclude the mass emailing, the response rate is substantially higher (closer to 33%).

presence of senior managers is useful because the views of these decision makers are particularly relevant.

About half of the respondents work for asset managers, 21% for mutual funds, 12% for pension funds, and a smaller number for hedge funds (4%). Our respondents tend to work for large investors, which is beneficial for our analysis because these investors are most likely to have the resources for and interest in shareholder engagement. Specifically, a total of 34 respondents—35% of the sample—work for institutions with assets under management of more than \$100 billion. Given that only 128 institutional investors had more than that amount in assets under management by the end of 2012 (Towers Watson (2013)), our survey provides a good representation of these very large investors' views.

The respondents represent a number of different regions of the world: 36% are from Continental Europe, 24% from the United States, 16% from the United Kingdom, and the rest are from other parts of the world.

We asked the investors for the typical holding period of the investments in their portfolios. Respondents could classify their holding periods into three categories: short (less than six months), medium (six months to two years), and long (more than two years). Table I, Panel B shows that, although none of the investors typically hold shares for less than six months, 29% have medium average holding periods and 71% have long average holding periods. The respondents hold most of their investments—75% on average—in active rather than purely passive (indexed) positions. Finally, we asked respondents to indicate the importance of stock liquidity when they consider buying or selling shares. Liquidity considerations are considered to be either somewhat important (53%) or very important (37%) by most of our respondents. Correlations across respondent characteristics are reported in Internet Appendix IA.IV. Given the responses, we are confident that in almost all cases we have only one observation per institutional investor. We should note that our respondent group is probably biased toward

¹⁰ The reason is that in 89% of all observations, we have sufficient responses to determine that none of the following identifying variables coincide: (i) location; (ii) assets under management; (iii) institutional investor type;

the more activist and long-term investors among the population of institutional investors. This outcome is the result of both the survey delivery methods and the fact that such investors would be more disposed to filling out such a survey. However, understanding the preferences of these investors is particularly important because they are more likely to shape the governance of firms. In addition, their activities may be a catalyst for other investors.

II. Shareholder Engagement Channels

A. Prevalence of Voice and Exit Channels

Traditional theoretical models of corporate governance focus on the benefits of direct intervention ("voice"), while more recent models focus on governance through exit. These theories raise the question of how widely voice and exit are actually contemplated or used by institutional investors. Addressing this question through archival research methods is problematic because many interventions may take place behind the scenes, making their observation and measurement difficult. Similarly, exit that results from dissatisfaction with a portfolio firm is empirically difficult to distinguish from exit resulting from other motives, such as liquidity or portfolio rebalancing needs. Thus, to shed light on these issues, respondents were asked to indicate what shareholder engagement measures they had taken in the past five years with their portfolio firms. To measure the importance of exit, investors could indicate two options: (i) whether they exited because of dissatisfaction with performance; and (ii) whether they exited because of dissatisfaction with corporate governance. We did not ask about exit resulting from other considerations unrelated to governance.

Table II reports the percentage of respondents that used a particular measure, as well as the results of *t*-tests that determine whether the percentage for a given measure is equal to the percentages of each of the other measures. We rank results based on their relative frequency.

⁽iv) investor horizon; (v) active investments (+/- 10% variation in the variable); and (vi) proxy advisor. (We explain this sixth variable below.) In the remaining 11%, we cannot exclude the possibility that respondents work for the same institutional investors, but the responses are sufficiently different to discount that possibility with some degree of assurance.

The findings indicate a generally very high level of engagement by our respondents, which, as discussed, may reflect our sample selection: Only 19% of the survey respondents have not taken any corrective actions in the past five years. The responses also indicate that investors use multiple channels to engage, relying primarily on voice, but also using exit. Discussions with management are the most frequently used engagement channel, with 63% of the respondents indicating that they have used it in the past five years. Moreover, 45% of the respondents conducted discussions with members of the board of directors *outside* of management's presence (the fourth most important response).

Our findings of the widespread use of private discussions support the view that investors first try to engage firms behind the scenes through direct negotiations, and take public measures (e.g., shareholder proposals, public criticism) only after the private interventions have failed. Our results might therefore help explain why many shareholder proposals filed by institutional investors are eventually withdrawn before the shareholder meeting, and why stock markets often do not react positively to shareholder proposals (e.g., Gillan and Starks (2000, 2007)).¹¹

Our results are consistent with the studies of individual institutions that engage behind the scenes with management (Carleton, Nelson, and Weisbach (1998), Becht et al. (2009), and Dimson, Karakas, and Li (2014)), and with Brav et al. (2008a) who document that many hedge funds (48% of their sample) are very active in seeking communication with management. Our results thus suggest that private discussions with management are not restricted to these particular investors or investor types, but rather are a more general phenomenon. We find that 53% of the respondents report voting against management as a shareholder engagement measure. This result is consistent with Iliev et al. (2015), who show that shareholder voting is an effective way through which corporate governance is exercised. Interestingly, submissions of shareholder proposals have only been used by 16% of the respondents. This low percentage

_

¹¹ A shareholder proposal is good news in that it indicates increased monitoring. But it also signals that institutional investors could not negotiate a behind-the-scenes agreement with management.

stands in stark contrast to the high frequency of behind-the-scenes discussions among our respondents. Two explanations for the infrequent use of shareholder proposals could be the rather low passage rates of proxy proposals (e.g., Gillan and Starks (2000, 2007)) and the fact that proxy fights are expensive (e.g., Gantchev (2013)). Moreover, if investors submit proposals only after discussions with management fail, this result would also suggest a rather high success rate for behind-the-scenes activities. In fact, representatives of both Vanguard and Blackrock have pointed out that they prefer engagement behind the scenes to the submission of shareholder proposals. Michelle Edkins, global head of corporate governance and responsible investment at Blackrock, is quoted as saying, "In our experience (private engagement) has a fair degree of traction with management. And we can raise (an) issue without having to dictate how management should address it. In a way, that's always the weakness of the shareholder proposal route" (Burr (2012)).

The responses in Table II also indicate that selling shares in portfolio firms because of dissatisfaction with performance or corporate governance (i.e., exit as a governance mechanism) is used by 49% and 39% of the respondents, respectively (56% of the investors have used one or the other exit mechanism).

Very extreme engagement channels also show relatively high usage, with 15% of the respondents having taken legal actions and 13% having publicly criticized their portfolio companies. These responses are consistent with recent evidence surrounding specific firms; for example, the buyout of Dell caused a number of institutional investors to publicly criticize the transaction and to take legal action (Benoit et al. (2013)). The overall observation that investors prefer private negotiations to public engagement is consistent with recent theoretical evidence in Levit (2014). In his model, if an activist's information becomes public, the activist loses credibility and the ability to influence the manager's actions.

B. Determinants of Voice Intensity

Theories of voice make different (often conflicting) predictions about the effects of investor or investment characteristics on the willingness to intervene. To examine the determinants of institutions' intensity of voice (where we consider intensity to reflect the different actions an investor takes), we construct a composite investor-specific "voice index". This index sums the different types of voice actions an investor has taken. Our survey has 11 possible intervention measures, implying that the voice index can vary between 0 and 11. Institutions with higher index numbers have shown a stronger tendency to engage along multiple dimensions. The average investor used three different types of interventions over the past five years.

In examining the potential determinants of an investor's voice intensity, we turn to the theoretical literature for guidance. The first characteristic we consider is the liquidity of a portfolio firm's stock, which researchers have argued is important for the intervention decision, although the direction of the effect is theoretically ambiguous. On the one hand, Coffee (1991) and Bhide (1993) argue that stock liquidity has a limiting effect because it encourages investors to "cut and run" rather than intervene. Back, Li, and Ljungqvist (2014) model this idea more formally and predict that liquidity harms activism. Similar predictions come from exit theories, which show that liquidity makes the exit threat more credible, reducing the need to govern through direct intervention (e.g., Edmans and Manso (2011)).

On the other hand, Faure-Grimaud and Gromb (2004) predict that higher stock liquidity results in more intervention. In their model, a shareholder is more likely to engage in monitoring when liquidity is higher. The reason is that liquidity makes it easier to exit at a price which reflects the engagement activities if the shareholder has to sell her stake prematurely. In addition, Maug (1998), Kahn and Winton (1998), and Collin-Dufresne and Fos (2014) show that liquidity facilitates block formation, which incentivizes intervention.

¹² For a more extensive review of the theories relating liquidity to shareholder intervention, see Edmans (2014).

Similar to liquidity, the relation between intervention and the horizon of the institutional investor has diverse predictions. Some argue that investors with longer horizons have stronger incentives to intervene. For example, such investors may be more interested in long-term profits; they may be more likely to remain shareholders of a firm long enough to realize the corresponding benefits; or they may have more time to understand and learn about a firm in order to intervene effectively (e.g., Burkart, Gromb, and Panunzi (1997), Faure-Grimaud and Gromb (2004), Gaspar, Massa, and Matos (2005), Chen, Harford, and Li (2007)). These views imply that investors with longer horizons intervene more often. However, others argue that activism is more often performed by short-term investors—in particular, hedge funds—who intervene in order to procure short-term profits. It is claimed that short-term investors push for actions that are profitable in the short term but are detrimental to firm value in the long term (e.g., Bratton and Wachter (2010)). These views imply that investors with shorter horizons intervene more often.

The size of an institutional investor can also be important for the use of voice. Larger investors are more likely to have larger stakes in their portfolio firms, and larger holdings provide stronger engagement incentives because they allow an investor to keep a larger share of the benefits if engagement is successful (Grossman and Hart (1980), Shleifer and Vishny (1986)). Larger funds generally also have more resources to engage.

Firms with large passive investments might also have stronger incentives to engage as they find it more difficult to use exit as a governance mechanism because of indexing or tracking error considerations. Alternatively, investors with large active holdings may trade more profitably on information collected through private engagement, which increases their incentives to engage (e.g., Maug (1998)). Finally, engagement might vary across institutions because of differences in regulation, compensation structure, expertise, or conflicts of interest. We therefore include investor type and geographical location as control variables.

Table III, Columns 1 to 5 report regressions that relate voice intensity to investor and investment characteristics. The results show that voice intensity is significantly negatively

related to the institutions' preferences for liquidity, which suggest that investors who care more about stock liquidity, and who arguably hold more liquid stocks, use voice less intensively. This implies that theories in which liquidity discourages voice are more strongly supported; liquidity either allows investors to cut and run, or it causes investors to use exit rather than voice. Our liquidity findings support empirical evidence on hedge fund activism showing that, conditional on block formation, liquidity discourages voice, which is measured using either 13D fillings (Edmans, Fang, and Zur (2013)) or activist campaigns (Back, Li, and Ljungqvist (2014)).

Our results also show that investors with longer holding periods use voice more intensively, possibly because their long-term orientation provides them with stronger incentives to monitor. This finding is inconsistent with the view that activism is primarily used by short-term—oriented investors, but it supports the arguments in Bebchuk, Brav, and Jiang (2014). In terms of economic significance, the ordered logit coefficient of 1.25 in Column 5 indicates that as the investor horizon changes from medium to long term, the odds of engaging along an additional voice dimension increase by 149%, holding the other variables constant. (In an OLS framework, we find that long-term investors use 1.6 more voice channels than medium-term investors.)

We also find weak evidence that investors with more active holdings use voice more intensively. Surprisingly, we find no evidence that intensive use of voice is confined only to large investors. However, because the investors in our sample are rather large, we do not want to over-interpret this result. Also, note that we interpret all regression coefficients as suggestive evidence with respect to the underlying theoretical arguments but not as tests of causality.

C. Voice and Exit: Substitutes or Complements

We next study whether exit and voice are used as complements or substitutes based on the information revealed by the investors. A number of authors argue that exit and voice are complements. For example, Hirschman (1970, p. 82) states that "the chances for voice to function effectively...are appreciably strengthened if voice is backed up by the threat of exit."

This idea has been modeled by theories that integrate exit and voice. For example, Edmans and Manso (2011) show that blockholders may engage both in intervention and exit. Levit (2014) shows that the option to exit improves the effectiveness of voice, even if a manager is not concerned about the short-term stock price. In his model, exit and voice are complements because the possibility of exiting improves the ability of an activist to influence the manager. Similarly, Dasgupta and Piacentino (2014) show that exit and voice can complement each other. In their model, voice comes in the form of costly shareholder proposals and managers only listen to shareholders if voice is backed up by a threat of exit. All three models predict that investors should rely both on exit and voice across portfolio firms.

Meanwhile, an argument also exists that exit and voice could be substitute measures, at least for some investors (e.g., Kahn and Winton (1998)). For example, some investors might lack expertise for intervention or face other impediments for engagement and, thus, rely on exit in their portfolios. Moreover, investors might face capital gains liabilities when exiting, which discourages exit but encourages voice (Jin (2006), Dimmock et al. (2014)).

To investigate these competing views, we create a dummy variable, which equals one if an institutional investor used exit as a governance mechanism over the past five years (because of dissatisfaction with corporate governance or performance). The regressions in Table III, Columns 6 and 7 relate the previously defined voice intensity index with this exit dummy. We find that both variables are positively and statistically significantly correlated, which means that investors that use exit as a governance mechanism also show a higher intensity of voice. This finding supports the theories that model exit and voice as complementary mechanisms.

III. The Threat of Exit

A. Measuring the Exit Threat

Exit models argue that blockholders can govern even if they do not actively intervene (e.g., Admati and Pfleiderer (2009), Edmans (2009), Edmans and Manso (2011)). The idea behind these models is that blockholders can collect private information on the fundamental value of a

firm, and this information can be impounded in the stock price through trading. If managers care about the stock price, for example because of equity-based pay or takeover fears, they want to avoid exit by informed blockholders. The threat of exit therefore induces managers to increase firm value, and exit will not take place if the exit threat is credible. Empirically, Parrino, Sias, and Starks (2003) show that exit can have beneficial effects because selling by institutional investors is associated with higher CEO turnover. Leuz, Lins, and Warnock (2009) find that foreign investors leave firms that do not improve governance.

A key challenge to testing exit models and evaluating the effectiveness of governancemotivated exit is that the threat of exit is unobservable. Our survey offers three alternative ways to address this challenge. First, we asked investors whether they engage management before they exit a firm because of their dissatisfaction with management. Knowing about such prior engagement is important because exit models assume that blockholders are informed about the fundamental value of a firm. Informed trading is more likely after prior engagement. Moreover, engagement prior to exit reinforces the exit threat by reminding management of the possibility that an informed investor will sell. Consistent with these models' assumptions, Table IV, Panel A shows that, prior to exiting, the large majority of investors first engage management (either sometimes (51%) or often (27%)). Second, we ask the investors whether they believe, based on their experience, that the threat of exit, rather than exit itself, causes management to change behavior (e.g., to take certain actions or perform better). The responses in Table IV, Panel B show that 42% believe that the threat of selling shares causes management to change behavior. Given that our respondents tend to be dedicated, longer-term investors that usually interact with management in private, it is plausible that their potential exit truly is a threat to firms.

Third, we try to understand the importance of investors' equity stake size for the threat of exit. Exit theories predict that stake size has two opposing effects. If stake size increases, an informed blockholder can sell more shares upon collecting negative information about the manager, and the incentives to collect information in the first place increase as well. Both

effects imply that the exit threat increases with stake size. However, if stake size becomes too large, selling the entire stake upon negative information becomes difficult because the price impact will be too large. This counter effect implies that an optimal stake size exists. Conditional on believing in the exit threat, we find in Table IV, Panel B that more than two-thirds of the respondents believe that the equity stake size should be at least 2% for the exit threat to be effective. Only a quarter of respondents think that the stake size should be beyond 10%. This relatively smaller number probably reflects that the exit threat is less credible if the stake is too large. However, these beliefs are not universal as 20% of respondents report that stake size does not matter, which implies that the exit threat can be effective even with small stakes. (The same effectiveness can be seen with relatively large ones, but institutional investors rarely hold stakes beyond 10%.)

Overall, these three results support the notion that the threat of exit is an important disciplinary governance mechanism, which is consistent with exit models.

B. Exit Threat: Determinants of Effectiveness

Theory predicts that the effectiveness of the exit threat depends not only on stake size, but also on a set of other factors. We use our survey in order to ask the investors to evaluate the importance of these factors. In addition to block size, other factors identified by exit models are a stock's liquidity, management's equity incentives, the existence of other blockholders (which could also be threatening to sell), investor flow concerns, and tracking error considerations.

Exit theories hypothesize that a stock's liquidity is central to the effectiveness of exit. For example, in Edmans (2009), liquidity increases the exit threat because it encourages blockholders to collect information on firm fundamentals; allows blockholders to trade more aggressively on that information; and leads to larger initial blocks. These positive effects counter a negative effect of liquidity, which is that liquidity reduces the price impact of a given blockholder's informed trade. Exit is also considered to be more credible if a firm's managers have greater equity ownership; managers then experience a larger loss if the stock price is

depressed because of a blockholder's exit. Edmans and Manso (2011) show that the exit threat is more effective if multiple informed blockholders hold shares in a firm because their trading then incorporates more information about the fundamental firm value into the stock price.

Some factors are hypothesized to weaken the threat of exit. For example, if institutional investors follow an index, they increase tracking error when they sell shares in a firm, which can make the exit threat less credible. In fact, tracking error considerations might even force investors to buy shares in situations where they would prefer to sell. Dasgupta and Piacentino (2014) show that the threat of exit can also be weaker if institutional investors care about investor flows (e.g., if the ultimate investors chase short-term performance). Thus, investors might not sell for fear that such an action makes their clients think that they initially bought the wrong stocks. In effect, the signaling role of exit then impairs its potential as a disciplinary threat.

To evaluate these factors, respondents were asked to indicate their importance for the effectiveness of the exit threat on a scale of 1 (not at all important) to 5 (very important). The results are provided in Table IV, Panel C. Investors state that the three most important factors influencing the effectiveness of an exit threat are: (i) selling by other investors for the same reason (72% consider this factor somewhat or very important); (ii) managerial equity ownership (70%); and (iii) the presence of large shareholders in the firm (67%). Interestingly, we find that even more subtle effects, such as investor flow considerations, are considered to be important by one out of four respondents. These findings support the mechanisms that have been highlighted in exit models.

IV. Impediments and Triggers to Shareholder Activism

A. Impediments to Shareholder Activism

Although some prominent intervention cases (e.g., Microsoft or Dell) have been highlighted recently in the news, some researchers argue that institutional investors too rarely initiate activism through voice (e.g., Black (1990)). In fact, there is an extended policy debate about

how to incentivize investors to be more active.¹³ This raises the question of why institutional investors would not be more active in pursuing changes in the companies in which they invest. While one explanation might be that investors govern more effectively through exit, additional reasons could be economic or legal impediments that hinder activism. Understanding these impediments is important for researchers trying to model the behavior of institutional investors, as well as for those who want to encourage more engagement.

A number of arguments have been made as to why investors do not engage more actively with their portfolio companies. These arguments focus on four potential areas: incentives to engage; conflicts of interest; legal barriers; and investment management industry structure. Grossman and Hart (1980) and Shleifer and Vishny (1986) argue that large investors face disincentives in becoming activists because of a "free rider" problem. That is, they would incur large costs from intervention, which would be borne solely by the activist, but any benefits from the activities would be spread among all shareholders. Theory predicts that this free rider problem decreases with stake size because a larger stake allows an investor to capture a bigger share of the value increase resulting from the intervention. Theory further predicts that the probability of successful intervention depends on the ownership structure of a firm, which can help or hinder intervention. For example, a larger number of blockholders reduces free rider problems and positively influences the effectiveness of voice (Winton (1993), Noe (2002), Edmans and Manso (2011)). However, management or insider control of voting rights reduces incentives to engage because such control decreases the probability that, for example, a proxy fight is successful.

Engagement might also be hindered by investors' conflicts of interest. Duan, Hotchkiss, and Jiao (2014) show that business ties provide valuable information to fund managers and can lead to profitable trades in the related stocks. Investors' concerns over existing or future business

-

¹³ See, for example, the Corporate Governance Green Paper issued by the European Commission (2011).

relations with the targeted firms also can inhibit intervention (Brickley, Lease, and Smith (1988), Cvijanovic, Dasgupta, and Zachariadis (2014)).¹⁴

In addition, investors might not intervene if they fear that they will breach legal rules. For example, diversification requirements for mutual funds or pension funds might not allow investors to take a stake that is sufficiently large to incentivize engagement. In addition, rules on "acting in concert" can discourage engagement because such rules imply a legal risk when investors coordinate engagement (e.g., the risk of violating Rule 13D in the U.S., or the risk of having to make a public offer in Europe). Finally, disclosure regulations (e.g., the U.S. "Regulation Fair Disclosure") might discourage investors or managers from engagement.

Apart from these factors, intervention can also be impeded because of the structure of the investment management industry. Fund managers might not engage if their own investors do not sufficiently reward activism or if the investment process is outsourced to other asset management firms. Further, some investors might not consider shareholder engagement to be part of their mandate, in which case they take a more passive stance. Finally, some investors might believe that corporate governance is not important to a firm's financial performance.

These arguments suggest that a number of impediments can prevent institutional investors from engaging a firm. To understand the importance of these impediments, we used two survey techniques. In one version of the survey, we asked respondents to indicate the importance of different factors on a scale of 1 (not at all important) to 5 (very important). In another version we asked respondents to mark on a list with the same factors their top four choices. The second technique circumvents the problem that many investors might indicate a high or very high importance for many different factors, making identification of the key mechanisms difficult.

The results for both techniques are provided in Table V. Consistent with the discussions in theory and practice, we don't find one single, very important reason for not engaging. This

18

¹⁴ However, Davis and Kim (2007) and Ashraf, Jayaraman, and Ryan (2012) find no evidence that business ties with portfolio firms influence voting.

finding means that academics wishing to model or measure investors' behavior must consider a range of impediments and their interplay. In addition, regulators who would like to encourage more engagement face a difficult challenge; they must address a range of economic and legal aspects because simple solutions do not seem to exist. Among the different hurdles, the most important ones identified are related to incentives: Respondents indicate they do not engage because the benefits from engagement are too small; their stakes in firms are too small; they have limited resources (personnel); and they have too many firms in their portfolios. These findings imply that free rider problems limit broad shareholder engagement, even for very large institutional investors like the ones in our respondent group.¹⁵

Investors also indicate that legal reasons, especially rules against concerted actions, discourage engagement. Thus, despite recent regulatory changes (Choi (2000)) and court rulings (e.g., CSX Corp. v. TCI case) that provide greater latitude for cooperation, legal concerns over coordinated engagement remain important. A further impediment is the belief that engagement makes receiving information from targeted firms more difficult; the implication is that conflicts of interest are deemed to be of some importance. Finally, our results indicate that the belief that corporate governance matters to firm performance is not universal: 19% of the respondents do not engage as they believe corporate governance does not affect performance.

B. Triggers of Shareholder Activism

We currently have little evidence on the actual triggers of shareholder activism. The reason is that engagement triggers are difficult to observe and need to be inferred from characteristics of targeted firms or information in SEC filings (e.g., Brav et al. (2008a)). To understand which firms are targeted, we asked respondents to indicate the importance of different potential intervention triggers. We again applied two techniques. In the first one, respondents were

_

 $^{^{15}}$ CalPERS, for example, focuses its activism only on the 300 firms of which it has the largest holdings (The Economist (2013)).

asked to indicate the level of importance of different triggers on a scale of 1 (not at all important) to 5 (very important). In the second one, they could indicate their top four triggers.

The results in Table VI show that the investors view a number of different triggers as important for engagement. Using the first technique, ten triggers receive an average score of 4 or higher. Consistently across both techniques, the main triggers—following the extreme case of fraud—are governance and strategy. More specifically, inadequate corporate governance and excessive compensation are considered by 88% of the respondents as somewhat or very important triggers, which reflects the importance of these topics in the public debate. Another important trigger for shareholder activism is disagreement with a firm's strategy. We find that 89% of the respondents consider poor corporate strategy and 82% large diversifying mergers or acquisitions as somewhat or very important triggers. However, the latter was named as a top-four trigger in only 5% of the responses under the second technique. Overall, these results indicate that investors not only engage over short-term issues (e.g., equity issues or low dividends) but also, and even more so, over long-run strategic issues. In the public debate is a strategy and second technique.

Consistent with a recent wave in shareholder proposals that ask companies to be more transparent about their political donations (Financial Times (2014)), we find that 40% of the respondents consider financial contributions to politicians as a somewhat or very important intervention trigger. Interestingly, dissatisfaction with company performance does not appear to be a key driver of shareholder engagement, although it is still considered to be somewhat important. This finding is consistent with Karpoff, Malatesta, and Walkling (1996), who study targets for shareholder proposals.

¹⁶ Interestingly, although fraud scores high as an intervention trigger when we use the first survey technique, it is not of first order importance when we use the second technique, which indicates that although fraud is an important concern, it is not expected to be a prevalent problem.

¹⁷ These results support Bebchuk, Brav, and Jiang (2014), who find no evidence that interventions by hedge funds are mostly motivated by short-term objectives. They are further consistent with Cornelli, Kominek, and Ljungqvist (2013), who find that managers are fired not because of incompetence or moral hazard, but because of a corporate strategy mismatch. They also line up with the finding in Brav et al. (2008a) that hedge fund activism targeting the sale of the company or changes in strategy is associated with the largest announcement returns.

V. Outsourced Shareholder Activism? The Role of Proxy Advisors

An important channel of shareholder activism is voting at the annual shareholder meeting (e.g., Iliev et al. (2015)). As we found earlier, 53% of the respondents have voted against management in their proxy votes as shareholders. Further, proxy voting itself is important for regulatory and fiduciary reasons, and some institutional investors are required (or voluntarily choose) to disclose their proxy votes, as well as their voting policies. However, proxy voting entails costs, particularly given that many investors have to cast votes on several thousand securities. As a result, institutional investors commonly rely on proxy advisors, at least for voting platforms in which the investors provide proxy advisors with instructions on how they want their shares voted. In addition, many institutions also rely on proxy advisors for information on which to base their vote. As a result, the proxy advisory industry has grown substantially during the past decade. The two major players in the U.S. are Institutional Shareholder Services (ISS) and Glass, Lewis & Co.

The current debate about proxy advisors centers around two issues: the quality of their voting advice and their conflicts of interest. Diverse views emerge in relation to the first issue. One view holds that proxy advisors are a reliable source of informed voting advice because they collect information, perform delegated monitoring, and use their expertise and experience to make informed voting recommendations (e.g., Ertimur, Ferri, and Oesch (2013)). Thus, the services of proxy advisors reduce investors' voting costs, leading to better and more informed voting decisions. The alternative view holds that the recommendations of proxy advisors are too standardized and ignore firm-specific circumstances (e.g., Gordon (2009)). Moreover, their recommendation criteria entail a lack of transparency, making the assessment of the quality of voting recommendations difficult. Both issues are a particular concern when investors use the advisors primarily to fulfill their duty to vote, but are not too concerned about the quality of their recommendations. In this case proxy advisors may have incentives to conduct only low-cost analyses (Larcker, McCall, and Ormazabal (2014)).

The second major issue of concern is whether proxy advisors have conflicts of interests in arriving at their recommendations. For example, ISS serves as a consultant, advising firms on how they can improve their corporate governance; at the same time, ISS makes recommendations about how investors in these firms should vote. Some researchers maintain that this dual role may give rise to recommendations that are affected by conflicts of interest (Yermack (2010), Alexander et al. (2010)). Counterarguments are offered by ISS, which argues that it has reduced conflicts of interest by structurally separating the voting advice from its consulting business. In addition, Li (2014) argues that the industry structure, consisting of only two main players, also reduces conflicts of interests.

These diverse views show that we need a better understanding of how investors view the role of proxy advisors. Our analysis proceeds in three steps. First, we examine the extent to which institutional investors use proxy advisors. Table VII, Panel A shows that 60% of our respondents use at least one proxy advisor, and almost half of these respondents employ the services of more than one advisor. Thus, many investors rely on more than one source of information when forming their views on voting matters.

Second, we study whether the use of proxy advisors substitutes or complements for governance through intervention. To this end, we create a dummy variable that equals one if an investor uses at least one proxy advisor, and zero otherwise. We then regress this variable on the previously defined voice index, which captures the intensity of intervention. The results are reported in Table VII, Panel B. We find that investors who employ the services of proxy advisors use direct intervention through voice more intensively, which implies that intervention and the services of proxy advisors are complements rather than substitutes. In terms of economic significance, the probit coefficient of 0.19 in Column 3 indicates that a one-unit increase in the voice index increases the odds of using a proxy advisor by 20%. The view that investors outsource activism to proxy advisors and then remain passive when it comes to engagement is thus refuted by these data.

Third, we address how institutional investors assess the role of proxy advisors in their own decision-making. In particular, we try to understand whether investors are worried that their recommendations are too standardized and reflect conflicts of interests, or whether they help them in making more informed voting decisions. These issues are difficult to answer using data on voting recommendations and votes because evaluating whether recommendations simply reflect investor preferences or actually influence them is challenging. The results are reported in Table VIII. We find that 55% of the respondents agree or strongly agree that proxy advisors help them make more informed voting decisions. Less than one-third of the respondents agree or strongly agree that their advice is too standardized. These results suggest that proxy advisors do not just aggregate shareholder preferences or coincide with them, but actually influence voting decisions in a positive way. These views are consistent with our finding that engagement and the use of proxy advisors are complements. They are also in line with Iliev and Lowry (2015) who find that most mutual funds take ISS recommendations into account but do not simply follow them, and Aggarwal, Erel, and Starks (2014) who find increasing divergence between proxy advisors and institutional investors in their voting choices over time.

The investors also raise some concerns about the work of proxy advisors; 52% agree or strongly agree that proxy advisors are sometimes exposed to conflicts of interest. Possibly as a result of this concern, 22% of the respondents agree or strongly agree that proxy advisors should be regulated.

VI. Conclusions

We use survey methodology to provide insights into the corporate governance preferences and actions of institutional investors. We focus on the two active choices investors have when they are unhappy with a firm: exit and voice. Guided by theoretical models and previous empirical evidence, we develop survey questions that directly assess institutional

-

¹⁸ As Michelle Edkins of Blackrock has stated in the media about proxy advisors, "They are really an important part of the process and help us identify the companies that we should be focusing our efforts on" (Burr (2012)).

investors' preferences and actions. Such knowledge can be captured only with a survey methodology. Thus, our survey allows us to validate existing theories and inferences made from indirect empirical tests of shareholders' preferences and actions regarding corporate governance.

Most importantly, through the survey we find institutional investors' preferences and actions to be largely consistent with the theoretical literature. Our survey shows that institutional investors frequently employ voice in their shareholder engagements. The most common engagement channels are behind-the-scenes discussions with management and boards of directors. Nevertheless, we find that investors still face several engagement impediments, principally because of liquidity concerns, free rider problems, and legal concerns. In addition, we show that investor horizon matters for engagement. First, long-term investors intervene more intensively than short-term investors. Second, engagements are primarily triggered by concerns over a firm's corporate governance or strategy rather than over short-term issues.

We then show that more than 40% of our respondents report that they believe the exit threat disciplines management. The effectiveness of this threat depends, according to our survey results, on investors' equity stake size, whether other investors also exit for the same reason, managerial equity ownership, and whether other large shareholders are also present. Moreover, exit and voice are related, as the surveyed investors believe that both governance mechanisms are complementary strategies.

Finally, we shed light on the controversial role of the proxy advisor, which has been a concern for many. Our survey responses indicate that proxy advisors do not just aggregate shareholder preferences or coincide with them, but actually influence voting in a positive way.

REFERENCES

Admati, Anat, and Paul Pfleiderer, 2009, The "Wall Street Walk" and shareholder activism: exit as a form of voice, *Review of Financial Studies* 22, 2245-2485.

Admati, Anat, Paul Pfleiderer, and Josef Zechner, 1994, Large shareholder activism and financial market equilibrium, *Journal of Political Economy* 102, 1097-1130.

Aggarwal, Reena, Isil Erel, and Laura Starks, 2014, Influence of public opinion on investor voting and proxy advisors, Working paper, University of Texas at Austin.

Alexander, Cindy, Mark Chen, Duane Seppi, and Chester Spatt, 2012, Interim news and the role of proxy voting advice, *Review of Financial Studies* 23, 4419-4454.

Back, Kerry, Tao Li, and Alexander Ljungqvist, 2014, Liquidity and governance, Working paper, NYU Stern.

Bebchuk, Lucian, Alon Brav, and Wei Jiang, 2014, The long-term effects of hedge fund activism, *Columbia Law Review*, forthcoming.

Becht Marco, Julian Franks, Colin Mayer, and Stefano Rossi, 2009, Returns to shareholder activism: Evidence from a clinical study of the Hermes UK Focus Fund, *Review of Financial Studies* 22, 3093-3129.

Benoit, David, Sharon Terlep, Joann S. Lublin, and Ben Worthen, 2013, Dell battles to protect deal—Founder defends \$24.4 billion buyout as large investor balks and rival circles, *Wall Street Journal*, March 7, A1.

Bhide, Amar, 1993, The hidden costs of stock market liquidity, *Journal of Financial Economics* 34, 31-51.

Black, Bernard, 1990, Shareholder passivity reexamined, Michigan Law Review 89, 520-608.

Bolton, Patrick, and Ernst-Ludwig von Thadden, 1998, Blocks, liquidity, and corporate control, *Journal of Finance* 53, 1-25.

Bratton, William, and Michael Wachter, 2010, The case against shareholder empowerment, *University of Pennsylvania Law Review* 158, 653-728.

Brav, Alon, Wei Jiang, Frank Partnoy, and Randall Thomas, 2008a, Hedge fund activism, corporate governance, and firm performance, *Journal of Finance* 63, 1729-1775.

Brav, Alon, John Graham, Campbell Harvey, and Roni Michaely, 2008b, Managerial response to the May 2003 dividend tax cut, *Financial Management* 27, 611-624

Brav, Alon, Wei Jiang, and Hyunseob Kim, 2010, Hedge fund activism: A review, *Foundations* and *Trends in Finance* 4, 185-246.

Brav, Alon, Wei Jiang, and Hyunseob Kim, 2015, The real effects of hedge fund activism: Productivity, asset allocation, and labor outcomes, Working paper, Duke University.

Brav, Alon, Wei Jiang, Song Ma, and Xuan Tian, 2014, Shareholder power and corporate innovation: Evidence from hedge fund activism, Working paper, Duke University.

Brickley, James, Ron Lease, and Clifford Smith, 1988, Ownership structure and voting on antitakeover amendments, *Journal of Financial Economics* 20, 267-292.

Burkart, Mike and Amil Dasgupta, 2014, Activist funds, leverage, and procyclicality, Working paper, Stockholm School of Economics and London School of Economics.

Burkart Mike, Denis Gromb, and Fausto Panunzi, 1997, Large shareholders, monitoring, and the value of the firm, *Quarterly Journal of Economics* 112, 693-728.

Burr, Barry, 2012, Money managers increasing activism on governance—but quietly, *Pensions and Investments* http://www.pionline.com/article/20120319/PRINT/303199980/money-managers-increasing-activism-on-governance-8212-but-quietly.

Carleton, Willard, James Nelson, and Michael Weisbach, 1998, The influence of institutions on corporate governance through private negotiations: Evidence from TIAA-CREF, *Journal of Finance* 53, 1335-1362.

Chen, Xia, Jarrod Harford, and Kai Li, 2007, Monitoring: Which institutions matter?, *Journal of Financial Economics* 86, 279-305.

Choi, Stephen, 2000, Proxy issue proposals: Impact of the 1992 SEC proxy reforms, *Journal of Law, Economics, & Organization* 16, 233-268.

Coffee, John C., 1991, Liquidity versus control: The institutional investor as corporate monitor, *Columbia Law Review* 91, 1277-368.

Collin-Dufresne, Pierre, and Vyacheslav Fos, 2014, Moral hazard, informed trading, and stock prices, Working paper, University of Illinois.

Cornelli, Francesca, Zbigniew Kominek, and Alexander Ljungqvist, 2013, Monitoring managers: Does it matter? *Journal of Finance* 68, 431-481.

Cvijanovic, Dragana, Amil Dasgupta, and Konstantinos Zachariadis, 2014, Ties that bind: How business connections affect mutual fund activism, Working paper, London School of Economics.

Dasgupta, Amil, and Giorgia Piacentino, 2014, The Wall Street Walk when blockholders compete for flows, *Journal of Finance*, forthcoming.

Davis, Gerald, and E. Han Kim, 2007, Business ties and proxy voting by mutual funds, *Journal of Financial Economics* 85, 552-570.

Dichev, Ilia, John Graham, Campbell Harvey, and Shiva Rajgopal, 2013, Earnings quality: Evidence from the field, *Journal of Accounting and Economics* 56, 1-33.

Dimmock, Stephen, William Gerken, Zoran Ivkovich, and Scott Weisbenner, 2014, Capital gains lock-in and governance choices, Working paper, University of Illinois at Urbana-Champaign.

Dimson, Elroy, Oguzhan Karakas, and Xi Li, 2014, Active ownership, Working paper, London Business School.

Duan, Ying, Edith Hotchkiss, and Yawen Jiao, 2014, Business ties and information advantage: Evidence from mutual fund trading, Working paper, Boston College.

Edmans, Alex, 2009, Blockholder trading, market efficiency, and managerial myopia, *Journal of Finance* 22, 4881-4917.

Edmans, Alex, 2014, Blockholders and corporate governance, *Annual Review of Financial Economics* 6, 23-50.

Edmans, Alex, and Gustavo Manso, 2011, Governance through trading and intervention: A theory of multiple blockholders, *Review of Financial Studies* 24, 2395-2428.

Edmans, Alex, Vivian Fang, and Emanuel Zur, 2013, The effect of liquidity on governance, *Review of Financial Studies* 26, 1443-1482.

Ertimur, Yonca, Fabrizio Ferri, and David Oesch, 2013, Shareholder votes and proxy advisors: Evidence from say on pay, *Journal of Accounting Research* 51, 951-996.

European Commission, 2011, Green paper: The EU corporate governance framework, Brussels.

Faure-Grimaud, Antoine, and Denis Gromb, 2004, Public trading and private incentives, *Review of Financial Studies* 17, 985-1014.

Financial Times, 2014, US companies pushed on political funds, June 22.

Gallagher, Daniel, 2013, Remarks at Society of Corporate Secretaries & Governance Professionals, 67th National Conference, Seattle, Washington (http://www.sec.gov/News/Speech/Detail/Speech/1370539700301#. U4xO4dy4nII).

Gantchev, Nickolay, 2013, The costs of shareholder activism: Evidence from a sequential decision model, *Journal of Financial Economics* 107, 610-631.

Gantchev, Nickolay, Oleg Gredil, and Chotibhak Jotikasthira, 2014, Governance under the gun: Spillover effects of hedge fund activism, Working paper, University of North Carolina.

Gaspar, Jose Miguel, Massimo Massa, and Pedro Matos, 2005, Shareholder investment horizons and the market for corporate control, *Journal of Financial Economics* 76, 135-165.

Gillan, Stuart, and Laura Starks, 2000, Corporate governance proposals and shareholder activism: The role of institutional investors, *Journal of Financial Economics* 57, 275-305.

Gillan, Stuart, and Laura Starks, 2007, The evolution of shareholder activism in the United States, *Journal of Applied Corporate Finance* 19, 55-73.

Gordon, Jeffrey, 2009, "Say on pay": Cautionary notes on the U.K. experience and the case for shareholders opt-in, *Harvard Journal on Legislation* 46, 323-64.

Grossman, Sanford, and Oliver Hart, 1980, Takeover bids, the free-rider problem, and the theory of the corporation, *Bell Journal of Economics* 11, 42-64.

Hirschman, Albert, 1970, Exit, Voice and Loyalty: Responses to Decline in Firms, Organizations, and States, Cambridge, MA, Harvard University Press.

Huddart, Steven, 1993, The effect of a large shareholder on corporate value, *Management Science* 39, 1407-1421.

Iliev, Peter, Karl Lins, Darius Miller, and Lukas Roth, 2015, Shareholder voting and corporate governance around the world, *Review of Financial Studies*, forthcoming.

Iliev, Peter, and Michelle Lowry, 2015, Are mutual funds active voters? *Review of Financial Studies* 28, 446-485.

Jin, Li, 2006, Capital gains tax overhang and price pressure, Journal of Finance 61, 1399-1431.

Kahn, Charles, and Andrew Winton, 1998, Ownership structure, speculation, and shareholder intervention, *Journal of Finance* 53, 99-129.

Karpoff, Jonathan, Paul Malatesta, and Ralph Walkling, 1996, Corporate governance and shareholder initiatives: Empirical evidence, *Journal of Financial Economics* 42, 365-95.

Larcker, David, Allan L. McCall, and Gaizka Ormazabal, 2014, Outsourcing shareholder voting to proxy advisory firms, Working paper, Stanford University.

Leuz, Christian, Karl Lins, and Frank Warnock, 2009, Do foreigners invest less in poorly governed firms? *Review of Financial Studies* 22, 3245-3285.

Levit, Doron, 2014, Soft shareholder activism, Working paper, Wharton School.

Li, Tao, 2014, Outsourcing corporate governance: Conflicts of interest and competition in the proxy advisory industry, Working paper, University of Warwick.

Maug, Ernst, 1998, Large shareholders and monitors: Is there a trade-off between liquidity and control?, *Journal of Finance* 53, 65-98.

Noe, Thomas, 2002, Investor activism and financial market structure, *Review of Financial Studies* 15, 289-318.

Parrino, Robert, Richard Sias, and Laura Starks, 2003, Voting with their feet: Institutional ownership changes around CEO turnovers, *Journal of Financial Economics* 68, 3-46.

Shleifer, Andrei, and Robert Vishny, 1986, Large shareholders and corporate control, *Journal of Political Economy* 94, 461-488.

Smith, Michael, 1996, Shareholder activism by institutional investors: Evidence from CalPERS, *Journal of Finance* 51, 227-252.

The Economist, 2013, Shareholders at the gates, March 9.

Towers Watson, 2013, The World's 500 largest asset managers, Industry Study.

Winton Andrew, 1993, Limitation of liability and the ownership structure of the firm, *Journal of Finance* 48, 487-512.

Yermack, David, 2010, Shareholder voting and corporate governance, *Annual Review of Financial Economics* 2, 103-125.

Table I Summary Statistics on Respondents

Panel A contains information on the characteristics of the institutional investors for which the respondents work. We report the respondents' position or job title, institutional investor types, assets under management, and location. Panel B contains information on the characteristics of the investments of the institutional investors. These characteristics include investor horizon, investment structure (percentage of the portfolio that is actively, rather than passively, invested), and stock liquidity. We obtained 143 responses to our survey, but not all respondents provided information on investor or investment characteristics.

Panel A: Institutional Investor Characteristics

Position of Respondent (N=97)	# Respondents	% Respondents
Corporate Governance or Proxy Voting Specialist	28	29%
Portfolio Manager	26	27%
Chief Investment Officer	17	18%
Board Member	14	14%
Other	12	12%
Institutional Investor Type (N=100)	# Respondents	% Respondents
Asset Manager	48	48%
Mutual Fund	21	21%
Pension Fund	12	12%
Hedge Fund	4	4%
Other	15	15%
Assets under Management (N=98)	# Respondents	% Respondents
More than \$100bn	34	35%
Between \$1bn and \$100bn	31	32%
Between \$100m and \$1bn	25	26%
Less than \$100m	8	8%
Location (N=102)	# Respondents	% Respondents
Continental Europe	37	36%
US	24	24%
UK	16	16%
Rest of World	25	25%

Panel B: Investment Characteristics

Investor Horizon (N=95)	# Respondents	% Respondents
Short (less than 6 months)	0	0%
Medium (6 months to 2 years)	28	29%
Long (more than 2 years)	67	71%
Investment Structure (N=91)	Mean	Median
Active Investments	75%	90%
Stock Liquidity (N=103)	# Respondents	% Respondents
Not at all important	2	2%
Somewhat unimportant	4	4%
Neither important nor unimportant	4	4%
Somewhat important	55	53%
Very important	38	37%

Table II Prevalence of Exit and Voice Channels

Respondents were asked to indicate what measures they have taken in the past five years with their portfolio companies. Responses were not mutually exclusive. Column 3 reports the percent of respondents that took a certain measure. Column 4 reports the number of respondents. Column 5 reports the results of a *t*-test of the null hypothesis that the percentage for a given measure is equal to the percentage for each of the other measures, where only significant differences at the 5% level are reported.

Statement	Shareholder engagement measures taken in the past five years:	% that	#	Significant
		took this	Respondents	differences
		measure		in Mean
				Response
				vs. Rows
(1)	Discussions with top management	63%	142	2-14
(2)	Voting against management	53%	142	1, 4-14
(3)	Selling shares because of dissatisfaction with performance	49%	142	1, 5-14
(4)	Discussions with board of directors outside of management	45%	142	1-2, 6-14
(5)	Selling shares because of dissatisfaction with corporate governance	39%	142	1-3, 8-14
(6)	Proposing a specific action to management	35%	142	1-4, 8-14
(7)	Aggressively questioning management on a conference call	30%	142	1-4, 8-13
(8)	Criticizing management and the board at the annual meeting	18%	142	1-7, 13
(9)	Publicizing a dissenting vote	18%	142	1-7, 13
(10)	Submitting shareholder proposals for the proxy statement	16%	142	1-7, 13
(11)	Legal action against management	15%	142	1-7, 13
(12)	Publicly criticizing management in the media	13%	142	1-7, 13
(13)	Changing SEC filings from Schedule 13G to 13D	1%	142	1-12, 14
(14)	None	19%	142	1-6, 13

Table III Determinants of Voice Intensity

This table reports ordered logit regressions. The dependent variable, *Voice*, is an index that counts the number of different voice channels an institutional investor has taken in the past five years with portfolio companies. The variable can vary between 0 and 11. *Stock Liquidity* is a variable that indicates the importance of the float or overall liquidity of shares when an institutional investor considers buying or selling shares. The variable ranges from 1 (not at all important) to 5 (very important). *Investor Horizon* is a variable that indicates the reported typical holding periods for investments in the portfolio of an institutional investor. The variable takes the values 0 (6 months to 2 years) or 1 (more than 2 years). *Assets under Management* is a variable that indicates the size of an institutional investor. The variable takes the values 1 (less than \$100m), 2 (between \$100m and \$1bn), 3 (between \$1bn and \$100bn), and 4 (more than \$100bn). *Active Investments* is a variable that measures the percentage of an institutional investor's portfolio that is invested actively. *Exit* is a dummy variable that equals one if an institutional investor used exit as a governance mechanism in the past five years with portfolio companies. We further include location and institutional investor type dummies. *t*-statistics, calculated based on robust standard errors, are reported in parentheses. ***, ***, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

				Voice			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Stock Liquidity	-0.39**				-0.54**		-0.61**
	(-2.24)	a a a strategic			(-2.11)		(-2.39)
Investor Horizon		1.34***			1.25**		1.30**
		(3.26)	0.45		(2.19)		(2.33)
Assets under Management			0.15		-0.09		-0.10
			(0.71)	0.04	(-0.30)		(-0.31)
Active Investments				0.01	0.02*		0.01
				(1.34)	(1.79)	a — a di di di	(1.59)
Exit						1.70***	1.05*
						(4.87)	(1.86)
US					0.33		0.61
					(0.44)		(0.78)
UK					1.73**		2.01**
					(2.18)		(2.16)
Continental Europe					-0.08		-0.04
					(-0.12)		(-0.07)
Hedge Fund					1.35		1.38
					(0.94)		(0.86)
Mutual Fund					1.27*		0.70
					(1.84)		(0.83)
Asset Manager					0.82		0.43
					(1.18)		(0.54)
Pension Fund					2.12***		1.79**
					(2.65)		(2.07)
N	103	94	97	90	85	142	85
Pseudo R ²	0.009	0.026	0.001	0.005	0.089	0.047	0.100

Table IV Effectiveness of the Threat of Exit

Panel A reports how often respondents engage management in an attempt to achieve changes *prior* to exiting due to dissatisfaction with management performance. Panel B reports whether respondents think that the *threat* of selling shares, rather than exit itself, causes management to make changes. If the answer was "Yes", they could further indicate how large the equity stake in a portfolio firm would need to be at minimum for the threat of exit to cause management to make such changes. In Panel C respondents indicate the level of importance of different situations for the effectiveness of the threat of exit on a scale of 1 (not at all important) to 5 (very important). Column 3 reports the mean score, where higher values correspond to higher importance. Column 4 presents the percent of respondents indicating importance levels of 4 or 5 (somewhat important or very important). Column 5 reports the number of respondents. Column 6 reports the results of a *t*-test of the null hypothesis that the mean score for a given situation is equal to the mean score for each of the other situations, where only significant differences at the 5% level are reported. Column 7 reports the results of a *t*-test of the null hypothesis that each mean score is equal to 3 (neither important nor unimportant). ****, ***, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

Panel A: Engagement Prior to Exit

Engagement of Management Prior to Exit?							
Often	Sometimes	Never	#				
			Respondents				
27%	51%	22%	89				

Panel B: Exit Threat: Effectiveness

Is the Threat of Exit Effective?									
No	34%								
Yes	42%	If "Yes": Minimum Stake Size?							
		Does not	At least	At least	At least	At least	#		
		matter	0.5%	2%	5%	10%	Respondents		
		20%	7%	22%	27%	24%	45		
Don't know	24%								
# Respondents	105								

Panel C: Exit Threat: Determinants of Effectiveness

Statement	Importance for the threat of exit to be effective:	Mean	% with	#	Significant	H ₀ : Mean
		Score	4 or 5	Respondents	differences	Score = 3
			Score		in Mean	
					Score vs.	
					Rows	
(1)	Selling of other investors for the same reason	3.87	72%	86	4-6	***
(2)	Equity ownership of the firm's management	3.76	70%	87	4-6	***
(3)	Existence of large shareholders in the firm	3.69	67%	88	5-6	***
(4)	Possibility to sell shares without affecting the price	3.40	56%	87	1-2, 5-6	***
(5)	Inference by clients about own stock picking ability	2.52	25%	87	1-4	***
(6)	Keeping shares to minimize tracking error	2.50	21%	84	1-4	***

Table V
Impediments to Shareholder Activism

This table reports survey responses based on two different techniques. Survey Technique A asked respondents to indicate the level of importance of different reasons for not conducting shareholder engagement on a scale of 1 (not at all important) to 5 (very important). Column 3 reports the mean score, where higher values correspond to higher importance. Column 4 presents the percent of respondents indicating importance levels of 4 or 5 (somewhat important or very important). Column 5 reports the number of respondents. Column 6 reports the results of a *t*-test of the null hypothesis that the mean score for a given reason is equal to the mean score for each of the other reasons, where only significant differences at the 5% level are reported. Column 7 reports the results of a *t*-test of the null hypothesis that each mean score is equal to 3 (neither important nor unimportant). Survey Technique B asked respondents to indicate the top four reasons for not conducting shareholder engagement. Column 8 reports the percent of respondents indicating a given reason as a top four reason for not conducting shareholder engagement. Column 9 reports the number of respondents. ***, **, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

Statement	Reasons for not conducting shareholder engagement:			Survey Techniqu	ie A		Survey	Survey Technique B		
		Mean	% with	#	Significant	H ₀ :	% Top 4	#		
		Score	4 or 5	Respondents	differences	Mean	Reason	Respondents		
			Score		in Mean	Score				
					Score vs.	= 3				
					Rows					
(1)	Benefits from engagement not large enough	3.4	61%	74	7-14	***	24%	41		
(2)	Too small of a stake in a firm	3.4	61%	74	7-14	**	51%	41		
(3)	Limited personnel	3.4	54%	74	7-14	**	56%	41		
(4)	Rules on "acting in concert" discourage coordination	3.3	44%	72	7-14	*	24%	41		
(5)	Too many firms in our portfolio	3.3	45%	75	7-14		27%	41		
(6)	Management or insider control of voting rights	3.2	45%	74	7-14		27%	41		
(7)	Investors in our fund do not sufficiently reward engagement	2.8	36%	74	1-6, 12-14		17%	41		
(8)	Disclosure regulations discourage conversations	2.8	25%	73	1-6, 12-14		7%	41		
(9)	Holdings by other institutional investors are not large enough	2.8	32%	74	1-6, 13-14	*	17%	41		
(10)	Engagement is not considered part of our investment mandate	2.7	36%	73	1-6, 13-14	*	20%	41		
(11)	Engagement makes it more difficult to receive information	2.6	28%	72	1-6, 14	**	22%	41		
(12)	Regulation does not allow us to take a sufficiently large stake	2.5	22%	74	1-8	***	7%	41		
(13)	Investment process is outsourced to other asset management firms	2.3	21%	73	1-10	***	10%	41		
(14)	Corporate governance does not affect financial performance	2.2	19%	72	1-11	***	10%	41		

Table VI
Triggers of Shareholder Activism

This table reports survey responses based on two different techniques. Survey Technique A asked respondents to indicate the level of importance of different triggers of shareholder engagement on a scale of 1 (not at all important) to 5 (very important). Column 3 reports the mean score, where higher values correspond to higher importance. Column 4 presents the percent of respondents indicating importance levels of 4 or 5 (somewhat important or very important). Column 5 reports the number of respondents. Column 6 reports the results of a *t*-test of the null hypothesis that the mean score for a given trigger is equal to the mean score for each of the other triggers, where only significant differences at the 5% level are reported. Column 7 reports the results of a *t*-test of the null hypothesis that each mean score is equal to 3 (neither important nor unimportant). Survey Technique B asked respondents to indicate the top four triggers for shareholder engagement. Column 8 reports the percent of respondents indicating a given trigger as a top four trigger for shareholder engagement. Column 9 reports the number of respondents. ***, **, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

Statement	Triggers for shareholder engagement:			Survey Techn	ique A		Survey	Survey Technique B		
		Mean	% with	#	Significant	H ₀ :	% Top 4	#		
		Score	4 or 5	Respondents	differences	Mean	Triggers	Respondents		
			Score		in Mean	Score				
					Score vs.	= 3				
					Rows					
(1)	Corporate fraud	4.5	89%	57	4-17	***	32%	37		
(2)	Inadequate corporate governance	4.4	88%	56	6, 8-17	***	81%	37		
(3)	Excessive management compensation	4.4	88%	56	8-17	***	41%	37		
(4)	Poor corporate strategy	4.2	89%	56	1, 11-17	***	43%	37		
(5)	Large diversifying merger or acquisition	4.1	82%	56	1, 12-17	***	5%	37		
(6)	Poor absolute financial performance	4.1	80%	56	1-2, 11-17	***	22%	37		
(7)	Poor financial performance relative to peers	4.1	79%	56	1, 12-13, 15-17	***	41%	37		
(8)	Large related-party transaction by insiders	4.1	79%	56	1-3, 15-17	***	30%	37		
(9)	Socially "irresponsible" corporate behavior	4.1	72%	57	1-3, 14-17	***	35%	37		
(10)	Large equity issuance	4.0	82%	55	1-3, 12-13, 15-17	***	0%	37		
(11)	Large negative earnings surprise	3.8	68%	56	1-4, 6, 16-17	***	5%	37		
(12)	Uncooperative management	3.7	64%	56	1-7, 10, 16-17	***	8%	37		
(13)	Suboptimal capital structure	3.7	68%	56	1-7, 10, 16-17	***	19%	37		
(14)	Earnings restatement	3.7	68%	56	1-6, 10, 16-17	***	3%	37		
(15)	Low payments to shareholders despite high cash holdings	3.7	71%	55	1-10, 16-17	***	19%	37		
(16)	Financial contributions to political parties or politicians	3.2	40%	55	1-15		5%	37		
(17)	The threat of a major shareholder to sell shares	2.8	27%	55	1-15		5%	37		

Table VII Role of Proxy Advisors

Panel A provides answers to the question of whether the respondents use proxy advisors. Panel B provides probit regressions of the determinants of their use. The dependent variable, *Proxy Advisor*, is a dummy that equals 1 if an institutional investor uses at least one proxy advisor and 0 otherwise. *Voice* is an index that counts the number of voice channels that an institutional investor had taken in the past five years with portfolio companies. The variable can vary between 0 and 11. *Investor Horizon* is a variable that indicates the reported typical holding periods for investments in the portfolio of an institutional investor. The variable takes the values 0 (6 months to 2 years) or 1 (more than 2 years). *Assets under Management* is a variable that indicates the size of an institutional investor. The variable takes the values 1 (less than \$100m), 2 (between \$100m and \$1bn), 3 (between \$1bn and \$100bn), and 4 (more than \$100bn). *Active Investments* is a variable that measures the percentage of an institutional investor's portfolio that is invested actively. We further include location and institutional investor type dummies. *t*-statistics, calculated based on robust standard errors, are reported in parentheses. ***, **, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

Panel A: Use of Proxy Advisors

# Proxy Advisor (N=99)	# Respondents	% Respondents
None	39	39%
One	32	32%
More than one	28	28%

Panel B: Determinants

		Proxy Advisor	
	(1)	(2)	(3)
Voice	0.27***		0.19**
	(4.97)		(2.49)
Investor Horizon		1.20***	0.97**
		(2.84)	(2.11)
Assets under Management		0.30	0.36*
		(1.35)	(1.65)
Active Investments		0.01	0.00
		(0.94)	(0.62)
US		0.59	0.61
		(1.03)	(1.03)
UK		1.45**	1.02
		(2.15)	(1.44)
Continental Europe		-0.66	-0.82
		(-1.46)	(-1.52)
Hedge Fund		1.80**	1.46*
		(2.07)	(1.72)
Mutual Fund		1.42***	0.91
		(2.59)	(1.59)
Asset Manager		1.74***	1.45**
		(3.28)	(2.44)
Pension Fund		2.65***	2.30**
		(3.24)	(2.57)
Constant	-0.68***	-3.41***	-3.58***
	(-2.91)	(-3.47)	(-3.48)
N	98	85	84
Pseudo R ²	0.198	0.416	0.459

Table VIII Assessment of Proxy Advisors

Respondents were asked to indicate the level of agreement with different statements about proxy advisors on a scale of 1 (strongly disagree) to 7 (strongly agree). Column 3 reports the mean score, where higher values correspond to higher agreement. Column 4 presents the percent of respondents indicating agreement levels of 6 or 7 (agree or strongly agree). Column 5 reports the number of respondents. Column 6 reports the results of a *t*-test of the null hypothesis that the mean score for a given statement is equal to the mean score for each of the other statements, where only significant differences at the 5% level are reported. Column 7 reports the results of a *t*-test of the null hypothesis that each mean score is equal to 4 (neither agree nor disagree). ***, **, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

Statement	Proxy advisors:	Mean	% with	#	Significant	H ₀ : Mean
		Score	6 or 7	Respondents	differences	Score = 4
			Score		in Mean	
					Score vs.	
					Rows	
(1)	Are sometimes exposed to conflicts of interest	5.29	52%	95	3-4	***
(2)	Allow us to make more informed voting decisions	5.22	55%	96	3-4	***
(3)	Offer too standardized advice	4.50	30%	96	1-2, 4	***
(4)	Should be regulated	4.06	22%	95	1-3	

Internet Appendix IA.I Survey on Shareholder Engagement, Shareholder Voting, and Proxy Voting Advisors

Intro	duction		
unde	are a team of professors from [XXX]. This survey is prestanding of the views and actions of institutional inveds of corporations in their investment portfolios.		
indiv	take the confidentiality of your responses very seriously idual firms or respondents be identified. Only aggregate by responses to any other data.		
-	\boldsymbol{u} would like to comment on any question in the survey, $\boldsymbol{\rho}$ urvey.	olease	do so at the space near the question or at the end of
If you	have any questions, please contact us: [XXX] or [XXX].		
Pleas	e send the survey back to one of our email addresses or t	o the	following postal address:
Than	[XXX] k you very much for participating in this survey.	K]	
Part	A: Shareholder Engagement		
	What measures of shareholder engagement have you tak panies? (Choose all that apply)	en <u>in</u>	the past five years with any of your portfolio
	None		Submitting shareholder proposals for the proxy statement
	Selling shares because of dissatisfaction with		
	performance		Discussions with members of the board of directors outside of management
	Selling shares because of dissatisfaction with corporate governance practices		Discussions with top management
	governance practices		Discussions with top management
	Voting against management at the annual meeting		Publicizing a dissenting vote
	Aggressively questioning management on a conference call		Criticizing management and the board at the annual meeting
	Publicly criticizing management in the media		Changing SEC filings from Schedule 13g ("passive ownership") to 13d ("active ownership") for U.S.
	Legal action against management (e.g., file a lawsuit, participate in a class action, file a "books and records"		investments
	request)		Proposing a specific action to management (e.g., sell assets, fire the CEO, increase payout, stop equity
	Other (please explain):		offering) (please also indicate the proposed action):

 Corporate governance does not affect financial performance		Shareholder engagement is not considered part of our investment mandate
 Expected benefits from shareholder engagement are not large enough to cover costs/expenses		Limited personnel to achieve shareholder engagement
 Too small of a stake in a firm		Our investment process is outsourced to other asset management firms
 Too many firms in our portfolio		-
 Holdings by other institutional investors are not large enough for sufficient pressure to get the desired results		Management or insider control of voting rights in the company is too large to allow achievement of the desired results
Disclosure regulations such as "Regulation Fair Disclosure" discourage conversations between management and investors		Rules on "acting in concert" discourage coordinating shareholder engagement with other institutional investors
 Regulation does not allow us to take a sufficiently large stake in a company		Shareholder engagement would potentially make it more difficult to receive information from company management
 Investors in our fund/institution do not sufficiently reward shareholder engagement		Other (please explain):
hich of the following corporate events or policies do ement? (Please rank #1, #2, #3, #4)	you view	as the top <u>four</u> triggers for shareholder
 Poor absolute financial performance		Poor financial performance relative to peers
 Poor absolute financial performance Poor corporate strategy	_	Poor financial performance relative to peers Large equity issuance
 Poor corporate strategy	_ _ _	Large equity issuance
 Poor corporate strategy Excessive management compensation	_ _ _ _	Large equity issuance Large diversifying merger or acquisition
 Poor corporate strategy Excessive management compensation Earnings restatement Large negative earnings surprise Financial contributions to political parties or	_ _ _ _ _	Large equity issuance Large diversifying merger or acquisition Suboptimal capital structure
 Poor corporate strategy Excessive management compensation Earnings restatement Large negative earnings surprise Financial contributions to political parties or politicians Low payments to shareholders despite high cash	 	Large equity issuance Large diversifying merger or acquisition Suboptimal capital structure The threat of a major shareholder to sell shares
 Poor corporate strategy Excessive management compensation Earnings restatement Large negative earnings surprise Financial contributions to political parties or politicians	 	Large equity issuance Large diversifying merger or acquisition Suboptimal capital structure The threat of a major shareholder to sell shares Corporate fraud Socially "irresponsible" corporate behavior (e.g., human rights violations or pollution) Large related-party transaction by insiders (e.g., sale of equity by management or sale of company
 Poor corporate strategy Excessive management compensation Earnings restatement Large negative earnings surprise Financial contributions to political parties or politicians Low payments to shareholders despite high cash holdings		Large equity issuance Large diversifying merger or acquisition Suboptimal capital structure The threat of a major shareholder to sell shares Corporate fraud Socially "irresponsible" corporate behavior (e.g., human rights violations or pollution) Large related-party transaction by insiders (e.g.,

Part B: Selling Shares Because of Dissatisfaction with Management Performance

B1: When you consider buying or selling shares, how important is the float or overall liquidity of the shares? (Check the
appropriate box)	

Not at all important		Somewhat	Neither important nor	Somewhat important	Very important			
	•	unimportant	unimportant	·	, ,			
	1	2	3	4	5			
B2: Did y	ou ever sell shai	res in companies, that	t is, exit the company's sto	ck, due to dissatisfaction	with management			
performa	nce?							
	No							
	Yes							
If your ar	swer was "ves"	· Refere you sold the	shares, how often did you	first try to angage mana	gement in an attemnt			
-	e changes?	. Before you sold the	silares, flow often did you	inst try to engage mana	gement in an attempt			
	_							
	Often							
	C +!							
	Sometimes							
	Never	evnlain):						
	Never	explain):						
□ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □	Never Other (please		es, rather than the exit its	elf, causes management	to make changes such			
-	Never Other (please ou think that the		-	elf, causes management	to make changes such			
-	Never Other (please ou think that the	threat of selling shar	-	elf, causes management	to make changes such			
as taking	Never Other (please ou think that the certain actions	threat of selling shar or performing better?	,	·	-			
as taking	Never Other (please ou think that the certain actions	threat of selling shar or performing better? equity stake in a portfo	olio firm is high enough (ple	·	-			

B4: How important do you consider the following situations to be in whether the threat of exit would be an effective disciplining mechanism? (Check the appropriate box)

Not at all important	Somewhat unimportant	Neither important nor unimportant	Somewhat important	Very important
1	2	3	4	5
1. The equity ownership a	and stock options of the	he firm's management tean	n	
2. The possibility to sell th	ne shares in the marke	et without affecting the pric	ce	
3. Whether other investor	rs are selling shares ir	the firm for the same reas	on	
4. The existence of large s	shareholders in the fir	·m		
5. Keeping some shares to	o minimize tracking ei	rror (but reducing exposure)	
6. An institutional investo investor picked the wrong	= -	n might be reluctant to exit	that position because clier	nts may infer that the

7. 0	ther (please e	xplain):					
							
Dart	: C: Proxy Voti	ng Advisors					
Part	. C. Proxy Voti	ilig Auvisors					
C2:	Yes, ju Yes, m No	n external proxy ist one nore than one nt do you agree		owing statem	ents regarding proxy	voting advisors	? (Check the
	Strongly	Disagree		Neither agree	Somewhat	Agree	Strongly
	disagree 1	2	disagree 3	nor disagree 4	agree 5	6	agree 7
	1	2	3	4	3	U	/
1. Pı	roxy voting ad	visors allow us to	o make more inform	ed voting deci	sions		
2.0							
2. PI	roxy voting ad	visors are somet	imes exposed to cor	ifficts of intere	ests if they offer cons	uiting services to	TITMS
3. Pı	roxy voting ad	visors should be	regulated				
		roxy voting advis	ors is too standardiz	ed and does r	ot sufficiently take fi	rm-specific circu	mstances into
acco							
Part	: D: General In	formation					
D1:	The fund/inst	itution where I	work can be best de	scribed as:			
	Hedge fund				Pension fund		
	Insurance co	ompany			Private equity fund		
	Mutual fund	l management co	ompany		Endowment, charity		
		ger for pension f ate accounts	unds, endowments a	and \square	Bank		
	Other (pleas	se specify):					
D2:	Approximate	y what percenta	age of your portfolio	is invested a	tively versus passive	ely?	
		ctive investment assive investmer					
D3:	What is the to	otal size of asset	s under managemer	nt for your org	anization?		
	Less than \$1	.00 million			Between \$1 billion a	ınd \$100 billion	
	Between \$1	00 million and \$	1 billion		More than \$100 bill	on	

	Short (less than 6 months)	
	Medium (6 months to 2 years)	
	Long (more than 2 years)	
D5: In wl	hat country are your headquarters based?	
D6: How	v would you best describe your principal investment style?	
	v would you best describe your principal investment style?at is your position?	
	at is your position?	
	at is your position? Chief Investment Officer	

Internet Appendix IA.II Survey Design

We developed an initial survey instrument by considering which questions would provide insights into important outstanding issues in corporate governance research. After designing an initial survey, we solicited feedback from 16 academics in finance and law and six practitioners (chief investment officers and corporate governance specialists at institutional investors). We also obtained feedback from two survey research experts on the design, wording, and execution of the survey.

The feedback from this group was sequential, so we were able to revise the survey multiple times and conduct beta tests to ensure that the questions would be properly understood and that the time required to complete the survey would be reasonable. Through this process, we adjusted the survey and deleted some questions. The final survey questions are provided in Internet Appendix IA.I.

The survey was designed to cover different aspects of corporate governance in three areas: Shareholder Engagement; Exit as a Governance Mechanism; and Proxy Voting Advisors. A final part of the survey was intended to gather investor demographic data. The survey also contained a section on Say on Pay, which is not covered in this paper because of space considerations. We did not require that institutional investors fill in all parts of the survey, and respondents could skip questions if they did not want to answer them.

We used both a paper version and an online version. The online version consisted of a series of linked webpages. To mitigate potential order-of-presentation effects, we randomly varied the order of choices within questions where we were concerned that such effects could be important (e.g., questions A1, A2, A3). Although we did not randomly order the response choices for the paper version of the survey, we did use two versions with two different orders of choices. We did not scramble the response choices for other questions where order appeared not to be an issue (e.g., for investor characteristics) or where a natural order of responses was logical (e.g., questions B2 and B3).

Internet Appendix IA.III compares the characteristics of the respondents to the online versus the paper versions of the survey. The appendix shows that continental European investors and mutual funds were the more likely respondents for the online survey, while pension funds were the more likely respondents for the paper version of the survey.

For two questions that aim at understanding the importance of different engagement impediments and triggers, we varied the survey design between two different techniques. In one version of the survey, we tried to discern the level of importance of different factors (e.g., reasons for not engaging) by asking respondents to indicate the importance of each factor on a scale of 1 (not at all important) to 5 (very important). In another version we asked respondents to mark on a list with the same factors their top four choices (e.g., top four reasons for not engaging). This survey version is included in the Internet Appendix IA.I. We included this second technique to enhance the validity of the responses and to circumvent the problem that many investors might indicate a high or very high importance for many different factors, making identification of the key mechanisms difficult.

Internet Appendix IA.III Respondent Characteristics: Online versus Paper Survey

This table compares the characteristics of the respondents to the online and paper versions of the survey, respectively. *Investor Horizon* is a variable that indicates the reported typical holding periods for investments in the portfolio of an institutional investor. The variable takes the values 0 (6 months to 2 years) or 1 (more than 2 years). *Stock Liquidity* is a variable that indicates the importance of the float or overall liquidity of shares when an institutional investor considers buying or selling shares. The variable ranges from 1 (not at all important) to 5 (very important). *Active Investments* is a variable that measures the percentage of an institutional investor's portfolio that is invested actively. *Assets under Management* is a variable that indicates the size of an institutional investor. The variable takes the values 1 (less than \$100m), 2 (between \$100m and \$1bn), 3 (between \$1bn and \$100bn), and 4 (more than \$100bn). *Proxy Advisor* is a dummy variable that equals 1 if an investor uses at least one proxy advisor and 0 otherwise. We further report significance levels of t-tests of the null hypothesis that the mean values of the characteristics of the online and paper respondents are equal to each other. ***, ***, and * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

	Onl	ine	Par	per	Difference
•	Mean	Obs.	Mean	Obs.	Significant?
Investor Horizon	0.7	58	0.7	37	
Stock Liquidity	4.3	66	4.0	37	**
Active Investments	79%	54	70%	37	
Assets under Management	3.0	59	2.8	39	
US	24%	59	23%	43	
UK	12%	59	21%	43	
Continental Europe	46%	59	23%	43	**
Hedge Fund	2%	61	8%	39	
Mutual Fund	30%	61	8%	39	***
Asset Manager	43%	61	56%	39	
Pension Fund	4%	99	18%	44	***
Proxy Advisor	55%	60	69%	39	

Internet Appendix IA.IV Correlations of Investor and Investment Characteristics

This table reports correlations between investor and investment characteristics. Assets under Management is a variable that indicates the size of an institutional investor. The variable takes the values 1 (less than \$100m), 2 (between \$100m and \$1bn), 3 (between \$1bn and \$100bn), and 4 (more than \$100bn). Active Investments is a variable that measures the percentage of an institutional investor's portfolio that is invested actively. Stock Liquidity is a variable that indicates the importance of the float or overall liquidity of shares when an institutional investor considers buying or selling shares. The variable ranges from 1 (not at all important) to 5 (very important). Proxy Advisor is a dummy variable that equals 1 if an investor uses at least one proxy advisor and 0 otherwise. * indicates significance at least at the 10% level.

	Assets under	Investor	Active	Stock Liquidity	Proxy Advisor
	Management	Horizon	Investments		
Investor Horizon	0.36*	1.00			
Active Investments	0.00	0.08	1.00		
Stock Liquidity	-0.01	-0.19*	0.23*	1.00	
Proxy Advisor	0.28*	0.33*	0.08	0.05	1.00
US	0.19*	-0.06	0.07	0.06	0.19*
UK	-0.11	0.15	0.16	0.02	0.30*
Continental Europe	0.12	0.10	-0.11	-0.23*	-0.32*
Rest of World	-0.25*	-0.19*	-0.10	0.19*	-0.10
Hedge Fund	-0.20*	-0.21*	-0.05	0.02	-0.05
Mutual Fund	0.22*	0.05	0.09	-0.13	0.05
Asset Manager	-0.01	-0.15	-0.14	0.08	0.10
Pension Fund	-0.01	0.18*	-0.08	0.04	0.24*